Author’s response to reviews

Title: Observational study of a new strategy and management policy for measles prevention in medical personnel in a hospital setting

Authors:

Chang-Pan Liu (joeliu5929@hotmail.com.tw)
Hsi-Peng Lu (lu@mail.ntust.edu.tw)
Tainyi Luor (dr.luor@gmail.com)

Version: 1 Date: 26 Oct 2018

Author’s response to reviews:

Dear editor,

Please find enclosed a manuscript entitled: "A new strategy and management policy for measles prevention in medical personnel" which I am submitting for consideration of publication as an article in BMC Infectious Diseases

The comments by the editor have been of great help in addressing deficiencies in the paper and are highly appreciated. We have studied your recommendations carefully. We have also carefully made point by point corrections for each of your comments. We hope that the manuscript meets the requirements for review processes in your esteemed Journal.

1. The need of consent from the subjects in this study was waived because this study was of the lowest risk, and the possible risk to the subjects participating in this study was not more than that for nonparticipants. It was approved by the Mackay Memorial Hospital Institutional Review Board. This study was within the scope of the conditions for which consent of study subjects was not required in research, according to national regulation, as stated in form 1010265083C issued by the Taiwan Ministry of Health and Welfare on the 5th of July 2012 to the institutional review board of medical organizations.

2. The corrections were made according to your recommendations.

3. We have checked the track changes or highlighting and submitted a clear version as your recommendations.
With best regards,

Sincerely yours,

*Corresponding author: Hsi-Peng Lu, Ph.D.

E-mail address: lu@mail.ntust.edu.tw / hsipeng@cs.ntust.edu.tw

Mailing address: Department of Information Management, National Taiwan University of Science and Technology, No. 43, Sec. 4, Keelung Road, Taipei, Taiwan.